Author's response to reviews

**Title:** High frequency of known copy number abnormalities and maternal duplication 15q11-q13 in patients with combined schizophrenia and epilepsy

**Authors:**

Larissa Stewart (larissa.stewart@gmail.com)  
April L. Hall (burant@bcm.edu)  
Sung-Hae L Kang (Sung-Hae.Kang@allina.com)  
Chad A Shaw (cashaw@bcm.edu)  
Arthur L Beaudet (abeaudet@bcm.edu)

**Version:** 4  **Date:** 5 October 2011

**Author's response to reviews:** see over
To the editor:

We are pleased to hear that our manuscript is now suitable for publication.

We have filled in the item where something was missing in the text in a revised version of that document which is attached. Dr. Levinson did respond and indicated that many of the same samples were studied with similar results. The data from Dr. Levinson's study are now accessible in dbGAP. In our manuscript, we have changed a few words to indicate the certainty that some of the same samples were studied.

Change in manuscript.

The publication by Levinson et al. [19] likely analyzed most of the same NIMH samples reported here.

Sincerely,

Arthur L. Beaudet